Author's response to reviews

Title: Are community midwives addressing the inequities in access to skilled birth attendance in Punjab, Pakistan? Gender, class and social exclusion

Authors:

Zubia Mumtaz (zubia.mumtaz@ualberta.ca)
Beverley O'Brien (beverley.obrien@ualberta.ca)
Afshan Bhatti (afshanbhattipk@yahoo.com)
Gian Jhangri (gian.jhangri@ualberta.ca)

Version: 2 Date: 4 August 2012

Author's response to reviews:

The Editor,
BMC Public Health
4th August 2012

Dear Sir,

Thank you for considering our submission titled 'Are community midwives addressing the inequities in access to skilled birth attendance in Punjab, Pakistan? Gender, class and social exclusion'.

In response to your request

1. Authors' information: Please include an Authors' Contributions section after Competing interests. Please check the instructions for authors on the journal website for the correct format to use for Authors' Contributions.

The following change was made on pg 23.

Author contributions

ZM, BB, AB conceptualized the study. ZM and BB prepared the manuscript. GJ provided statistical input. All authors approve the final submission.

Thank you for your time and consideration and we eagerly await your reply

Sincerely

Dr. Zubia Mumtaz, MBBS, MPH, PhD
Alberta Heritage Foundation for Medical Research Population Health Investigator
Assistant Professor, Global Health
School of Public Health
University of Alberta
3-309, Edmonton Clinic Health Academy
11405-87 Ave
Edmonton, Alberta T6G 2T4